

Respected Reviewer,

Comments: Dear author's I was pleased to review your editorial letter and i have some minor comments: A short abstract it will be informative. It is mandatory to add references in the first part of the letter. Please explain the novelty and clinical utility. Minor punctuation edits.

Reply: We are grateful for you time and constructive feedback of our manuscript.

- We have added a brief abstract to the manuscript.
- A reference has been added to the first paragraph.
- Novelty is highlighted throughout the paragraph and as such we have maintained it in its original form.

Sincerely,

Faisal S Ali, MD

Sushovan Guha, MD, PhD